\begin{thebibliography}{10}
\providecommand{\url}[1]{#1}
\def\UrlFont{\rmfamily}
\providecommand{\newblock}{\relax}
\providecommand{\bibinfo}[2]{#2}
\providecommand\BIBentrySTDinterwordspacing{\spaceskip=0pt\relax}
\providecommand\BIBentryALTinterwordstretchfactor{4}
\providecommand\BIBentryALTinterwordspacing{\spaceskip=\fontdimen2\font plus
\BIBentryALTinterwordstretchfactor\fontdimen3\font minus
  \fontdimen4\font\relax}
\providecommand\BIBforeignlanguage[2]{{%
\expandafter\ifx\csname l@#1\endcsname\relax
\typeout{** WARNING: IEEEtran.bst: No hyphenation pattern has been}%
\typeout{** loaded for the language `#1'. Using the pattern for}%
\typeout{** the default language instead.}%
\else
\language=\csname l@#1\endcsname
\fi
#2}}

\bibitem{sas}
``{SAS Institute},'' \url{www.sas.com/}.

\bibitem{BHH:Statistics}
G.~E.~P. Box, W.~G. Hunter, and J.~S. Hunter, \emph{Statistics for
  Experimenters}.\hskip 1em plus 0.5em minus 0.4em\relax New York: John Wiley
  \& Sons, 1978.

\bibitem{CART}
L.~Breiman, J.~Freidman, R.~Olshen, and C.~Stone, \emph{Classification and
  Regression Trees}.\hskip 1em plus 0.5em minus 0.4em\relax Monterey, CA:
  Wadsworth, 1984.

\bibitem{oats}
R.~Brownlie, J.~Prowse, and M.~S. Padke, ``Robust testing of {AT\&T
  PMX}/{S}tar{MAIL} using {OATS},'' \emph{{\em AT}\&{\em T Technical Journal}},
  vol.~71, no.~3, pp. 41--7, 1992.

\bibitem{burr}
K.~Burr and W.~Young, ``Combinatorial test techniques: Table-based automation,
  test generation and code coverage,'' in \emph{Proceedings of the Intl. Conf.
  on Software Testing Analysis \& Review}, 1998.

\bibitem{Childers:03}
B.~Childers, J.~Davidson, and M.~Soffa, ``{Continuous Compilation: A New
  Approach to Aggressive and Adaptive Code Transformation},'' in
  \emph{Proceedings of the International Parallel and Distributed Processing
  Symposium}, Apr. 2003.

\bibitem{dalal}
S.~R. Dalal, A.~Jain, N.~Karunanithi, J.~M. Leaton, C.~M. Lott, G.~C. Patton,
  and B.~M. Horowitz, ``Model-based testing in practice,'' in \emph{Proceedings
  of the Intl. Conf. on Software Engineering, (ICSE)}, 1999, pp. 285--294.

\bibitem{dunietz}
I.~S. Dunietz, W.~K. Ehrlich, B.~D. Szablak, C.~L. Mallows, and A.~Iannino,
  ``Applying design of experiments to software testing,'' in \emph{Proceedings
  of the Intl. Conf. on Software Engineering, (ICSE '97)}, 1997, pp. 205--215.

\bibitem{Grundy:01}
J.~Grundy, Y.~Cai, and A.~Liu, ``{Generation of Distributed System Test-beds
  from High-level Software Architecture Description},'' in \emph{{16 $^th$
  International Conference on Automated Software Engineering, Linz
  Austria}}.\hskip 1em plus 0.5em minus 0.4em\relax IEEE, Sept. 2001.

\bibitem{kola}
W.~Kolarik, \emph{Creating Quality: Systems, Concepts, Strategies and
  Tools}.\hskip 1em plus 0.5em minus 0.4em\relax McGraw-Hill Education, 1995.

\bibitem{Schmidt:05w}
A.~S. Krishna, A.~Gokhale, D.~C. Schmidt, V.~P. Ranganath, J.~H. White, and
  D.~C. Schmidt, ``{Model-driven Middleware Specialization Techniques for
  Software Product-line Architectures in Distributed Real-time and Embedded
  Systems},'' in \emph{Proceedings of the MODELS 2005 workshop on MDD for
  Software Product-lines}, Half Moon Bay, Jamaica, Oct. 2005.

\bibitem{Schmidt:04k}
A.~S. Krishna, D.~C. Schmidt, A.~Porter, A.~Memon, and D.~Sevilla-Ruiz,
  ``{Improving the Quality of Performance-intensive Software via
  Model-integrated Distributed Continuous Quality Assurance},'' in
  \emph{Proceedings of the 8th International Conference on Software
  Reuse}.\hskip 1em plus 0.5em minus 0.4em\relax Madrid, Spain: ACM/IEEE, July
  2004.

\bibitem{Schmidt:04n}
A.~S. Krishna, N.~Wang, B.~Natarajan, A.~Gokhale, D.~C. Schmidt, and G.~Thaker,
  ``{CCMPerf: A Benchmarking Tool for CORBA Component Model Implementations},''
  in \emph{Proceedings of the 10th Real-time Technology and Application
  Symposium (RTAS '04)}.\hskip 1em plus 0.5em minus 0.4em\relax Toronto, CA:
  IEEE, May 2004.

\bibitem{kuhn}
D.~Kuhn and M.~Reilly, ``An investigation of the applicability of design of
  experiments to software testing,'' \emph{Proceedings 27th Annual NASA
  Goddard/IEEE Software Engineering Workshop}, pp. 91--95, 2002.

\bibitem{GME:01}
A.~Ledeczi, A.~Bakay, M.~Maroti, P.~Volgysei, G.~Nordstrom, J.~Sprinkle, and
  G.~Karsai, ``{Composing Domain-Specific Design Environments},'' \emph{IEEE
  Computer}, pp. 44--51, Nov. 2001.

\bibitem{LiuY:02}
Y.~Liu, I.~Gorton, A.~Liu, N.~Jiang, and S.~Chen, ``{Designing a Test Suite for
  Empirically-based Middleware Performance Prediction},'' in \emph{{40 $^{th}$
  International Conference on Technology of Object-Oriented Languages and
  Systems, Sydney Australia}}.\hskip 1em plus 0.5em minus 0.4em\relax
  Australian Computer Society, Aug. 2002.

\bibitem{Lu-FCS:02}
C.~Lu, J.~A. Stankovic, G.~Tao, and S.~H. Son, ``{Feedback Control Real-Time
  Scheduling: Framework, Modeling, and Algorithms},'' \emph{Real-Time Systems
  Journal}, vol.~23, no. 1/2, pp. {85--126}, July 2002.

\bibitem{mandl}
R.~Mandl, ``Orthogonal {L}atin squares: an application of experiment design to
  compiler testing,'' \emph{Communications of the ACM}, vol.~28, no.~10, pp.
  1054--1058, 1985.

\bibitem{Schmidt:04b}
A.~Memon, A.~Porter, C.~Yilmaz, A.~Nagarajan, D.~C. Schmidt, and B.~Natarajan,
  ``{Skoll: Distributed Continuous Quality Assurance},'' in \emph{Proceedings
  of the 26th IEEE/ACM International Conference on Software Engineering}.\hskip
  1em plus 0.5em minus 0.4em\relax Edinburgh, Scotland: IEEE/ACM, May 2004.

\bibitem{Memon01c}
A.~M. Memon, M.~E. Pollack, and M.~L. Soffa, ``Hierarchical gui test case
  generation using automated planning,'' \emph{IEEE Transactions on Software
  Engineering}, vol.~27, no.~2, pp. 144--155, February 2001.

\bibitem{d-optimal}
T.~Mitchell, ``An algorithm for the construction of the 'd-optimal'
  experimental designs,'' \emph{Technometrics}, vol.~16, no.~2, pp. 203--210,
  1974.

\bibitem{sqc96}
D.~C. Montgomery, \emph{Introduction to Statistical Quality Control},
  3rd~ed.\hskip 1em plus 0.5em minus 0.4em\relax New York: John Wiley \& Sons,
  1996.

\bibitem{EMULab:03}
{Robert Ricci and Chris Alfred and Jay Lepreau}, ``{A Solver for the Network
  Testbed Mapping Problem},'' \emph{{SIGCOMM Computer Communications Review}},
  vol.~33, no.~2, pp. 30--44, Apr. 2003.

\bibitem{Schmidt:00a}
D.~C. Schmidt, M.~Stal, H.~Rohnert, and F.~Buschmann, \emph{{Pattern-Oriented
  Software Architecture: Patterns for Concurrent and Networked Objects, Volume
  2}}.\hskip 1em plus 0.5em minus 0.4em\relax New York: Wiley \& Sons, 2000.

\bibitem{tuk77}
J.~W. Tukey, \emph{Exploratory Data Analysis}.\hskip 1em plus 0.5em minus
  0.4em\relax Reading, Massachusetts: Addison-Wesley, 1977.

\bibitem{OCML_ACMSE:04}
E.~Turkay, A.~Gokhale, and B.~Natarajan, ``{Addressing the Middleware
  Configuration Challenges using Model-based Techniques},'' in
  \emph{Proceedings of the 42nd Annual Southeast Conference}.\hskip 1em plus
  0.5em minus 0.4em\relax Huntsville, AL: ACM, Apr. 2004.

\bibitem{White02}
B.~White and J.~L. et~al, ``{An Integrated Experimental Environment for
  Distributed Systems and Networks},'' in \emph{Proceedings of the Fifth
  Symposium on Operating Systems Design and Implementation}.\hskip 1em plus
  0.5em minus 0.4em\relax Boston, MA: USENIX Association, Dec. 2002, pp.
  255--270.

\bibitem{wu}
C.~F.~J. Wu and M.~Hamada, \emph{Experiments: Planning, Analysis, and Parameter
  Design Optimization}.\hskip 1em plus 0.5em minus 0.4em\relax Wiley, 2000.

\bibitem{cohenTSE06}
C.~Yilmaz, M.~B. Cohen, and A.~Porter, ``Covering arrays for efficient fault
  characterization in complex configuration spaces,'' \emph{IEEE Transactions
  on Software Engineering}, vol.~31, no.~1, pp. 20--34, Jan. 2006.

\bibitem{ca}
------, ``Covering arrays for efficient fault characterization in complex
  configuration spaces.'' in \emph{ISSTA}, 2004, pp. 45--54.

\bibitem{ICSE05}
C.~Yilmaz, A.~S. Krishna, A.~Memon, A.~Porter, D.~C. Schmidt, A.~Gokhale, and
  B.~Natarajan, ``Main effects screening: a distributed continuous quality
  assurance process for monitoring performance degradation in evolving software
  systems,'' in \emph{ICSE '05: Proceedings of the 27th international
  conference on Software engineering}.\hskip 1em plus 0.5em minus 0.4em\relax
  New York, NY, USA: ACM Press, 2005, pp. 293--302.

\bibitem{Padua:03}
K.~Yotov, X.~Li, G.~Ren, M.~Cibulskis, G.~DeJong, M.~Garzaran, D.~Padua,
  K.~Pingali, P.~Stodghill, and P.~Wu, ``{A Comparison of Empirical and
  Model-driven Optimization},'' in \emph{{Proceedings of ACM SIGPLAN conference
  on Programming Language Design and Implementation}}, June 2003.

\bibitem{zhang:02}
R.~Zhang, C.~Lu, T.~Abdelzaher, and J.~Stankovic, ``{Controlware: A Middleware
  Architecture for Feedback Control of Software Performance},'' in
  \emph{Proceedings of the International Conference on Distributed Systems
  2002}, July 2002.

\end{thebibliography}

% $Id: new_mes.tex,v 1.8 2006/07/24 02:36:27 aporter Exp $
\section{Performance-Oriented Regression Testing}
%%%\vspace*{-0.2in}
\label{mes}
   
As software systems evolve, developers often run regression tests to
detect unintended functional side effects.  Developers of
perfor\-mance-inten\-sive systems must also detect unintended side
effects on end-to-end QoS.  A common way to detect these effects is to
run benchmarks when the system changes.  As described in
Section~\ref{intro}, however, these efforts can be confounded for
systems with many possible configurations because time and resource
constraints (and often high change frequencies) se\-verely limit the
number of configurations that can be examined using only in-house
resources.

For example, our earlier experiences applying Skoll to
ACE+\-TAO~\cite{Schmidt:04b} showed that ACE+TAO developers have a
limited view of their software's QoS since they routinely benchmark
only a small number of common configurations.  QoS degradations not
readily seen in these configurations, therefore, can and do escape
detection until systems based on ACE+TAO are fielded by
end-users~\cite{Schmidt:04b,Schmidt:04k}.  The key problem here is
that the ACE+TAO developers are benchmarking a small and
unrepresentative sample of system configurations, so their
extrapolations from this data are bound to be unreliable. 

To address this problem we have developed and evaluated the {\em
reliable effects screening} process, which uses ``design of
experiments'' theory~\cite{kola} 
% Done: Adam, Atif, can you please add a citation for DoE here?
to determine an appropriate subset of system configurations to
benchmark when the system changes.  This section describes how we
implemented the reliable effects screening process, applied it to
ACE+\-TAO+\-CIAO, and disscusses several process choices developers must
make when applying it.

%%%\vspace*{-0.2in}
\subsection{The Reliable Effects Screening Process}
%%%\vspace*{-0.2in}
\label{mse}

Reliable effects screening (RES) is a process we developed to detect
performance degradation rapidly across a large configuration space as
a system changes.  This process identifies a small subset of
``important'' configuration options that substantially affect
variation in performance.  Benchmarking a ``screening suite''
containing all combinations of these important option settings (with
other options assigned randomly) should therefore provide a reliable
estimate of performance across the entire configuration space at a
fraction of the cost and effort of exhaustive benchmarking.

At a high level the process involves the following steps:
\begin{enumerate}
\item {\em Compute} a formal experimental design based on the system's
QA task model.

\item {\em Execute} that experimental design across volunteered
computing resources in the Skoll computing grid by running and
measuring benchmarks on specific configurations dictated by the
experimental design devised in Step 1.

\item {\em Collect, analyze and display} the data so that developers
can identify the most important options, {\em i.e.}, the options that
affect performance most significantly.

\item Conduct {\em supplementary analysis} again on volunteered
computing resources to check the basic assumptions underlying Step 1
and to confirm the results of Step 3.

\item {\em Estimate} overall performance (in-house, if possible)
whenever the software changes by evaluating all combinations of the
important options (while randomizing all other options).

\item Frequently {\em recalibrate} the important options by restarting
the process since these effects can change over time, depending on how
rapidly the subject system changes.
\end{enumerate}

\subsection{Screening Designs Background}
% Adam, Atif, I feel like this subsection is hard to follow since it
% it long and seems to combine a bunch of topics without adequate
% substructure.  Could you please add some \noindent {\bf ...}
% subheadings to make it easier to follow?

The first step of reliable effects screening is to identify options
accounting for the most performance variation across the system's QA
task space.  We do this by executing and analyzing formally-de\-signed
experiments, called {\em screening designs}, which are described in
Kolarik~\cite{kola}, Wu and Hamada~\cite{wu}, or the NIST Engineering
Statistics Handbook
(\url{www.itl.nist.gov/div898/handbook/index.htm}). Screening designs
are highly economical plans for identifying important low-order
effects, {\em i.e.}, first-, second-, or third-order effects (where an
$n^{th}$-order effect is an effect caused by the simultaneous
interaction of $n$ factors).

To better understand screening designs, consider a full factorial
({\em i.e.}, exhaustive) experimental design involving $k$ independent
binary factors.  The design's run size (number of experimental
observations) is therefore $2^k$.  Although such designs allow all
$1^{st}$- through $k^{th}$-order effects to be computed, they quickly
become computationally expensive to run.  Screening designs, in
contrast, reduce costs by observing only a carefully selected subset
of a full factorial design.  The tradeoff is that they cannot compute
most higher-order effects because the selection of observations
aliases the effects of some lower-order interactions with some
higher-order ones, {\em i.e.}, it conflates certain high- and
low-order effects.

Which effects are conflated depends on the design's {\em
resolution}. In resolution $R$ designs, no effects involving $i$
factors are aliased with effects involving less than $R-i$
factors. For instance, a resolution III design is useful to evaluate
``clear'' (no two aliased together) $1^{st}$-order effects, where all higher effects
are negligible.  Here all $1^{st}$-order effects will be clear, but
they may be aliased with $2^{nd}$- or higher-order effects.  This
conflation is justified only if the high-order effects are indeed
negligible.  If these assumptions are patently unreasonable, then a
higher resolution may be needed.


In practice, statistical packages are used to compute specific
screening designs. We used the SAS/QC~\cite{sas} package in this work,
but many other packages, such as MINITAB and SPSS, are also
applicable.  these packages will produce a screening design, assuming
one can be found, given the following
application-specific information: (1) a list of options and their
settings, (2) a maximum run size, and (3) the design's resolution.

% Done: Adam, Atif, yikes, the term ``configuration model'' comes out of the
% blue here!  Can you please provide a smoother transition from the
% previous paragraph(s) to this one?
Since we already build our QA task model graphically using Skoll's MDE
tools (see Section~\ref{skolloverview}), we can just use a translator
to convert it into the list of options and settings expected by our
statistical package.  The second and third items are interwined and
must be chosen by developers.  In particular, higher resolution designs
will yield more accurate estimates (assuming some higher-level effects
exist), but require more observations.  It is also often advisable to
run more than the minimum number of observations needed for a given
resolution to improve precision or to deal with noisy
processes. Developers must balance these competing forces.

% Adam, Atif, this might be a good place to add a new \noindent {\bf
% ... } subheading so that it would be easier for readers to follow
% the structure of the subsection?!

\subsection{Computing a Screening Design}

To demonstrate these choices, consider a hypothetical software system
with 4 independent binary configuration options, $A$ through $D$, each
with binary settings \verb|+| and \verb|-|.  A full factorial design
for this system involves 16 ($2^4$) observations.  We assume that our
developers can only afford to gather 8 observations. With so few
observations, there is no design with clear $1^{st}$- and
$2^{nd}$-order effects.  Developers must therefore either allow more
observations or limit themselves to capturing only the 4
$1^{st}$-order effects, {\em i.e.}, the effect of each option by
itself.  We assume they choose to stay with a run size of 8 and to use
a resolution IV design.

Given these choices, the developers generate one acceptable design
using a statistical package. The design (which appears in
Table~\ref{options}) is identified uniquely as a $2^{4-1}_{IV}$
design, which means that the total number of options is 4, that they
will observe a $1/2$ ($2^{-1}=1/2$) fraction of the full factorial
design, and that the design is a resolution IV screening design. 
\begin{table}[tbhp]
\centering
  \begin{tabular}{cccc}
    \hline
    A & B & C & D \\
    \hline\hline
    - & - & - & - \\
    + & - & - & + \\
    - & + & - & + \\
    + & + & - & - \\
    - & - & + & + \\
    + & - & + & - \\
    - & + & + & - \\
    + & + & + & + \\
    \hline
\end{tabular}
\caption{$2^{4-1}_{IV}$ Design (Binary Option Settings are Encoded as (-) or (+))}
\label{options}
\end{table}
The design tool also outputs the aliasing structure $D=ABC$. We can
see this aliasing in Table~\ref{options} where the setting of option
$D$ is the product of the settings of options $A$, $B$ and $C$ (think
of \verb|+| as 1 and \verb|-| as $-$1).  This dependence explains why
the effect of option $D$ cannot be untangled from the interaction
effect of options $A$, $B$, and $C$.

\subsection{Executing and Analyzing Screening Designs}

After defining the screening design, developers will execute it across
the computing resources comprising the Skoll grid.  In our later
feasibility studies, each experimental observation involves measuring
a developer-supplied benchmark program while the system runs in a
particular configuration.  Our QA engineers use BGML to generate
workload and benchmark code.  Once the data is collected we analyze it
to calculate the effects.  Since our screening designs are balanced
and orthogonal by construction ({\it i.e.}, no bias in the observed
data), the effect calculations are simple.  For binary options (with
settings \verb|-| or \verb|+|), the effect of option $A$, $ME(A)$, is
%%%\vspace*{-0.1in}
\begin{align}
ME(A) = z(A-) - z(A+)
\label{effect1}
\end{align}
where $z(A-)$ and $z(A+)$ are the mean values of the observed data
over all runs where option $A$ is (\verb|-|) and where option $A$ is
(\verb|+|), respectively.
 
If required, $2^{nd}$-order effects can be calculated in a similar
way.  The interaction effect of options $A$ and $B$, $INT(A, B)$ is:
%%%\vspace*{-0.1in}
\begin{align}
INT(A, B) &= 1/2 \{ME(B|A+) - ME(B|A-)\} \\
&= 1/2 \{ME(A|B+) - ME(A|B-)\}
\label{effect2}
\end{align}
Here $ME(B|A+)$ is called the {\em conditional effect} of $B$ at the
\verb|+| level of $A$.  The effect of one factor ({\em e.g.}, $B$)
therefore depends on the level of the other factor ({\em e.g.}, $A$).
Similar equations exist for higher order effects and for designs with
non-binary options. See Wu and Hamada~\cite{wu} for further details.
 
Once the effects are computed, developers will want to determine which
of them are important and which are not.  There are several ways to
determine this, including using standard hypothesis testing.  We do
not use formal hypothesis tests primarily because they require strong
assumptions about the standard deviation of the experimental samples.
Instead, we display the effects graphically and let developers use
their expert judgment to decide which effects they consider
important. While this approach has some downsides (see
Section~\ref{conclusions}), even with traditional tests for
statistical significance, experimenters must still judge for
themselves whether a significant effect has any practical importance.

Our graphical analysis uses {\em half-normal probability plots}, which
show each option's effect a\-gainst their corresponding coordinates on
% Adam, why are the subscripts for these thetas outside of the bars. I
% am not familiar with the notation so I am just curious - andy
the half-normal probability scale.  If $|\theta|_{1} \le |\theta|_{2}
\le ... \le |\theta|_{I}$ are the ordered set of effect estimations,
the half-normal plot then consists of the points
%%%\vspace*{-0.15in}
\begin{align}
 (\Phi^{-1}(0.5 + 0.5[i-0.5]/I), |\theta|_{i})\;for\;i=1,...,I
\end{align}
where $\Phi$ is the cumulative distribution function of a standard
normal random variable.

The rationale behind half-normal plots is that unimportant options
will have effects whose distribution is normal and centered near 0.
Important effects will also be normally distributed, but with means
different from 0.\footnote{Since the effects are averages over
numerous observations, the central limit theorem guarantees
normality.}  Options whose effects deviate substantially from 0 should
therefore be considered important.  If no effects are important, the
resulting plot will show a set of points on an approximate line near
$y=0$.

\subsection{Conduction Supplementary Analysis}

At this point developers have a candidate set of important
options. One potential problem, however, is that we arrived at these
options by making the following assumptions:
  
\begin{enumerate}
\item The low-order effects identified as being important 
  really are; while the higher-order effects they are aliased to are
  not.

\item Monitoring only low-order effects is sufficient to produce
reliable estimates. 
\end{enumerate}

Since these are only assumptions, it is important to check them before
proceeding since the reliability of our results will be severely
compromised if they do not hold.  We therefore validate these
assumptions using two types of follow-up experiments:
\begin{enumerate}
\item We first examine additional configurations to disambiguate any
effects aliased to our purported low-level important options.

\item We then examine additional configurations to look for other
higher-order effects.
\end{enumerate}
In these follow-up experiments we rely on another class of efficient
experimental designs called {\it D-optimal designs}~\cite{d-optimal},
which are again computer-aided designs.  Given a configuration space
and a model the experimenter wishes to fit, a D-optimal design uses
search-based computer algorithms ({\it e.g.}, hill climbing or
simulated annealing) to select a set of configurations that satisfy a
particular optimality criterion.  Unlike more common fractional
factorial designs, therefore, the size of D-optimal designs need not
be a perfect fraction of full factorial designs.  D-optimal designs
are preferable to standard classical designs when (1) the standard
designs require more observations than can be tested with available
time and resources and (2) the configuration space is heavily
constrained ({\it i.e.,} when not all the configurations are valid).
Both factors are frequently present in modern software systems. Full
details of D-optimail designs are beyond the scope of this paper, but
can be found in books and articles (See Kolarik~\cite{kola} and
Mitchell~\cite{d-optimal}).

Based on the results of this D-optimal design analysis, developers may
modify the set of important options. At this point developers have a
working set of important options that they can use to create a
screening suite of configurations to benchmark whenever the system
changes.
